Author's response to reviews

Title: An examination of knowledge, attitudes and practices related to lead exposure in South Western Nigeria

Authors:

Eugenious O Adebamowo (e_adebamowo@yahoo.com)
Oluwole A Agbede (oluwoleagbede@yahoo.com)
Mynepalli K Sridhar (mkcsridhar@yahoo.com)
Mynepalli K Sridhar (mkcsridhar@yahoo.com)
Clement A Adebamowo (cadebamo@yahoo.com)

Version: Date: 24 February 2006

Reviewer: JF Rosen MD

Reviewer's report:

General
We have deleted the reference to a normal lead level in page 3
The Grosse reference has been changed to Rothenberg and Rothenberg

We have improved the English used and where the verbatim report of the FGD contained grammatical errors, we have indicated this.

We have changed the formatting of the manuscript to conform to the style of the journal